Channel One - Unfiltered Simulated Underlying Signal and Path Effects

0 50 100 150 200 250

-5 0 5

Channel Two - Unfiltered Simulated Underlying Signal and Path Effects

0 50 100 150 200 250

-5 0 5 10

Channel Three - Unfiltered Simulated Underlying Signal and Path Effects

0 50 100 150 200 250

-5 0 5

Channel Four - Unfiltered Simulated Underlying Signal and Path Effects

0 50 100 150 200 250

-10 -5 0 5

Channel Five - Unfiltered Simulated Underlying Signal and Path Effects

0 50 100 150 200 250

-5 0 5 10